

Transcript Review for Secondary Social Studies: History Designation Certification

Students must maintain a 2.5 GPA in their teaching major and no course with a grade less than C- will be accepted for fulfillment of requirements.

U.S. History 4 courses	Term	Grade
1. HIST 210-1: History of the United States I	_____	
2. HIST 210-2: History of the United States II	_____	
3. _____	_____	
4. _____	_____	

European History 2 courses	Term	Grade
1. HIST 201-1: European Civilization I	_____	
2. HIST 201-2: European Civilization II	_____	

Non-western Civilizations (choose 2)

1. HIST 255-1, 2, 3: Background of African Civilization & Culture	
2. HIST 270-0: Islam in History	
3. HIST 281-0: Chinese Civilization	_____
4. HIST 284-0: Japanese Civilization	_____
5. HIST 285-0: Indian Civilization	
6. HIST 381-1, 2: History of Modern China	
7. HIST 384-1, 2: History of Modern Japan	
8. HIST 385-0: History of India	
9. HIST 365-0: Formation of Latin American Society	
10. HIST 366-0: Latin America in the Independence Era	
11. HIST 367-0: Politics and Development in Latin America	
12. HIST 368-1, 2: Revolution in 20 th Century Latin America	

Additional history courses to make up a total of 11 for WCAS or 12 for SESP students

1. _____	_____
2. _____	_____
3. _____	_____
4. _____	_____

Related core courses: (choose at least three although all cover state requirements.)

POL SCI 220-0: American Government	1. _____
SOC 110-0: Introduction to Sociology	2. _____
ECON 201-0: Macroeconomics	3. _____
ECON 202-0: Macroeconomics	
ANTHRO 214-0: Culture Origins	
SOC 226-0 or SESP 210-0 or STATS 210-0	
PSYCH 110-0: Introduction to Psychology	
GEOG 341-0: Principles of Cartography	